Author's response to reviews

Title: Prospective association between self-reported life satisfaction and mortality: Results from the MONICA/KORA Augsburg S3 Survey Cohort Study.

Authors:

Maria E Lacruz (lacruz@helmholtz-muenchen.de)
Rebecca T Emeny (rebecca.emeny@helmholtz-muenchen.de)
Jens Baumert (baumert@helmholtz-muenchen.de)
Karl H Ladwig (ladwig@helmholtz-muenchen.de)

Version: 5 Date: 6 July 2011

Author's response to reviews:

Thanks for the remark. Now it reads in page 17, paragraph 1:

"Indeed, only when either self-rated health or social network index were excluded from the psychological model, was the LS association with mortality restored (data not shown)."